Author's response to reviews

Title: Reduced genetic influence on childhood obesity in small for gestational age children

Authors:

Dug Yeo Han (dy.han@auckland.ac.nz)
Rinki Murphy (r.murphy@auckland.ac.nz)
Angharad R Morgan (ang.morgan@auckland.ac.nz)
Wen J Lam (w.lam@auckland.ac.nz)
John M.D Thompson (j.thompson@auckland.ac.nz)
Clare R Wall (c.wall@auckland.ac.nz)
Karen E Waldie (k.waldie@auckland.ac.nz)
Edwin A Mitchell (e.mitchell@auckland.ac.nz)
Lynnette R Ferguson (l.ferguson@auckland.ac.nz)

Version: 5 Date: 16 October 2012

Author's response to reviews: see over